Author's response to reviews

Title: Clinical implications of determination of safe surgical margins by using a combination of CT and 18FDG-positron emission tomography in soft tissue sarcoma.

Authors:

Mikio Terahara (teraharamikio@yahoo.co.jp)
Satoshi Nagano (naga@m2.kufm.kagoshima-u.ac.jp)
Yoshiya Arishima (yarisma@m3.kufm.kagoshima-u.ac.jp)
Michihisa Zemmyo (kumikohe@m2.kufm.kagoshima-u.ac.jp)
Masahiro Yokouchi (masahiro@m3.kufm.kagoshima-u.ac.jp)
Takako Yoshioka (yoshioka@m2.kufm.kagoshima-u.ac.jp)
Akihide Tanimoto (akit09@m3.kufm.kagoshima-u.ac.jp)
Setsuro komiya (skomiya@m3.kufm.kagoshima-u.ac.jp)

Version: 3 Date: 19 July 2011

Author's response to reviews: see over
Dear Editor;

We have checked our manuscript as the final proofing stage. The formatting changes are as follows.

**Major revisions.**
Conclusion: We included a “conclusions” section after the discussion.
Tables: We have removed the vertical lines in our tables.

**Minor revisions**
Patients and Methods: We renamed this section “Methods”.

We also changed the first author from MT to MY.

Thank you very much for all of your help.

Best wishes,

Masahiro Yokouchi, MD. PhD.
Department of Orthopaedic Surgery,
Kagoshima Graduate School of Medical and Dental Sciences, 8-35-1 Sakuragaoka, Kagoshima, Japan 890 - 8520
Tel. +81-99-275-5381
Fax. +81-99-265-4699
E-mail: masahiro@m3.kufm.kagoshima-u.ac.jp